Author's response to reviews

Title: Vacuum-assisted breast biopsy in close proximity to the skin: a case report.

Authors:
Flora Zagouri (florazagouri@yahoo.co.uk)
Theodoros N. Sergentanis (tsergentanis@yahoo.gr)
Dimitra Koulocheri (dimkoul@altecnet.gr)
Georgia Giannakopoulou (ggiannakopoulou@gmail.com)
Aphrodite Nonni (afnonni@med.uoa.gr)
Dimitrios Dardamanis (dardaman@hol.gr)
Nikolaos V. Michalopoulos (nmichal@med.uoa.gr)
Ioannis Flessas (iflessas@gmail.com)
John Bramis (breastunit_ipp@yahoo.gr)
George C. Zografos (gzografo@med.uoa.gr)

Version: 2 Date: 14 October 2007

Author's response to reviews: see over
Dear Editor,

I am sending for publication to your distinguished Journal the Case Report entitled: “Vacuum-assisted breast biopsy in close proximity to the skin: a case report.”.

According to your comments:

- The title page is kept to one page, and includes the title, authors, author affiliations and corresponding author details.
- The section "Background" has been renamed to "Introduction"
- A Consent section has been included after the Conclusions section.

Yours sincerely,

George C. Zografos,
Assoc. Professor of Surgery
University of Athens